 Vc/27-Sep-04
Week38 starting 13Sept  04

SM18 Operation Meeting (Tuesdays)

 Bench status  on Monday 13 Sept 04  ~15.30 hrs 

	TBA1
	MBAL3208
	ICS 3 Traveller after warm test
	299.1

	TBA2
	MBAL3203
	PT 12 Warm Up
	55.2

	TBB1
	_
	_
	298.6

	TBB2
	MBAR1140
	PT 12 Warm Up
	301.4

	TBC1
	MBAL3205
	PT 6.4 Training 4
	4.6

	TBC2
	MBBL2086
	Prep 5 Pumping + Cool Down
	104.2

	TBD1
	SSS043
	Finished Ready to go Final scan
	298.7

	TBD2
	SSS046
	Prep 5 Pumping + Cool Down
	298.8

	TBE1
	MBBL1132
	Prep 5 Pumping + Cool Down
	58.6

	TBE2
	MBBL1144
	PT 12 Warm Up
	13.4

	TBF1
	MBBL1143
	ICS 6 Disconnect Magnet WP07
	299

	TBF2
	MBBL1125
	PT 12 Warm Up
	296.9


9/13/2004 3:29:42 PM    Temperature data are taken from TBSM18 site
Coordinator this week: Marco; MTM support on call :  (16-3602).

RECORD !!  13 Magnets [ 9  dipoles( 3 with MM)+ 2 SSS]   Disconnected after tests last week ( Sat Midnight to Sat midnight)  
· Situation From Cryogenics:  J-P Lamboy or ACR Rep.
Priority settings …
1140 came back 3rd time ?

·  SSS work :   Wire Breakages
Nair + Arun leaving 

Back to Homepage
Tests 

	Magnets under Test from 05-09-2004 to 11-09-2004 

	-
	Bench
	Magnet
	Shafts
	Start Date .
	Connect
Time
	Cooling 
Time
	Cold Test 
Time
	Warming 
Time
	Disconnect
Time
	End Date
	Total
	
	

	1
	TBA2
	MBAL2047 
	[image: image1.png]



	31-08-2004 
	26 h.
	77 h.
	39 h.
	34 h.
	5 h.
	08-09-2004 
	192 hours
	
	

	2
	TBB1
	SSS035 
	[image: image2.png]



	30-08-2004 
	74 h.
	57 h.
	69 h.
	19 h.
	45 h.
	10-09-2004 
	264 hours
	
	

	3
	TBB2
	MBAL3206 
	[image: image3.png]



	06-09-2004 
	27 h.
	47 h.
	26 h.
	20 h.
	8 h.
	11-09-2004 
	120 hours
	
	

	4
	TBB2
	MBBL2107 
	[image: image4.png]



	30-08-2004 
	28 h.
	40 h.
	32 h.
	25 h.
	6 h.
	06-09-2004 
	168 hours
	
	

	5
	TBC1
	MBAL3145 
	[image: image5.png]



	02-09-2004 
	1 h.
	14 h.
	15 h.
	20 h.
	6 h.
	05-09-2004 
	72 hours
	
	

	6
	TBC1
	MBAL3202 
	[image: image6.png]



	05-09-2004 
	38 h.
	39 h.
	31 h.
	15 h.
	14 h.
	10-09-2004 
	120 hours
	
	

	7
	TBC2
	MBBL1099 
	[image: image7.png]



	05-09-2004 
	38 h.
	72 h.
	20 h.
	13 h.
	7 h.
	11-09-2004 
	144 hours
	
	

	8
	TBC2
	MBBL1146 
	[image: image8.png]



	27-08-2004 
	23 h.
	52 h.
	57 h.
	50 h.
	6 h.
	05-09-2004 
	216 hours
	
	

	9
	TBD2
	SSS038 
	[image: image9.png]



	24-08-2004 
	130 h.
	31 h.
	107 h.
	9 h.
	22 h.
	07-09-2004 
	336 hours
	
	

	10
	TBE1
	MBAL3159 
	[image: image10.png]



	02-09-2004 
	26 h.
	112 h.
	24 h.
	23 h.
	9 h.
	11-09-2004 
	216 hours
	
	

	11
	TBE2
	MBBL3195 
	[image: image11.png]



	01-09-2004 
	21 h.
	109 h.
	25 h.
	25 h.
	12 h.
	09-09-2004 
	192 hours
	
	

	12
	TBF1
	MBAR1133 
	[image: image12.png]



	29-08-2004 
	39 h.
	55 h.
	36 h.
	71 h.
	8 h.
	07-09-2004 
	216 hours
	
	

	13
	TBF2
	MBAL3146 
	[image: image13.png]



	31-08-2004 
	17 h.
	100 h.
	58 h.
	17 h.
	10 h.
	09-09-2004 
	216 hours
	
	


Quench Information
	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBAL2047
	11852 A.
(D1-U-S12)
	9201 A.
(No Data)
	12353 A.
(D1-L-S12)
	[image: image14.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	SSS035
	12064 A.
(QI-34)
	12850 A.
[image: image15.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	MBAL3206
	12011 A.
(D1-U)
	12850 A.
[image: image16.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	MBBL2107
	10347 A.
(D2-U)
	12539 A.
(D2-U)
	[image: image17.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	MBAL3145
	12580 A.
(D1_U)
	12655 A.
(D2-L)
	[image: image18.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	MBAL3202
	10082 A.
(D1-U)
	11583 A.
(D1-U)
	12096 A.
(D1-U)
	12654 A.
(D1_L)
	12850 A.
[image: image19.jpg]



	-
	-
	-
	-
	-
	-
	-
	-

	7
	MBBL1099
	12178 A.
(D2-U)
	12219 A.
(D1-U)
	[image: image20.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	8
	MBBL1146
	12461 A.
(D2-L)
	12462 A.
(D1 L)
	[image: image21.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	SSS038
	12661 A.
(QE_34)
	12856 A.
(QE-12)
	[image: image22.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	10
	MBAL3159
	10396 A.
(D2-U)
	12284 A.
(D2-U)
	[image: image23.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	11
	MBBL3195
	11521 A.
(D1-L)
	12596 A.
(D2_U)
	[image: image24.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	12
	MBAR1133
	12118 A.
(D1_U S03)
	12318 A.
(D1-L-S01)
	[image: image25.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	13
	MBAL3146
	11780 A.
(D2-U)
	11534 A.
(D1-U)
	11862 A.
(D2-L)
	12178 A.
(D2-U)
	12268 A.
(D1-U)
	12381 A.
(D2_L)
	12526 A.
(D1-L)
	12660 A.
(D2-U)
	12828 A.
(D1-U)
	[image: image26.png]



	-
	-
	-


Bench status  and tests for now (13/09/2004 15:33)  & next few days :  see Video Display

 

	Magnets under Test on 13-09 3:32:16 PM 

	-
	Bench
	Magnet
	Shafts
	Start
	Connect
Time
	Cooling 
Time
	Cold Test 
Time
	Warming 
Time
	Disconnect
Time
	Last Test
Desc
	
Time
	
I
	
Temp.
	Total

	1
	TBA1
	MBAL3208 
	[image: image27.png]



	13-09 
	4 h.[image: image28.png]



	-
	-
	-
	-
	ICS 3 Traveller after warm test
	13-09
(14 : 59) 
	0 A. 
	300 K
	4 hours

	2
	TBA2
	MBAL3203 
	[image: image29.png]



	08-09 
	22 h.
	64 h.
	16 h.
	21 h.[image: image30.png]



	-
	PT 12 Warm Up
	12-09
(18 : 54) 
	0 A. 
	Warming
	123 hours

	3
	TBB2
	MBAR1140 
	[image: image31.png]



	12-09 
	39 h.[image: image32.png]



	-
	-
	-
	-
	PT 12 Warm Up
	13-09
(12 : 41) 
	0 A. 
	Warming
	39 hours

	4
	TBC1
	MBAL3205 
	[image: image33.png]



	10-09 
	22 h.
	32 h.
	14 h.[image: image34.png]



	-
	-
	PT 6.4 Training 4
	13-09
(14 : 13) 
	0 A. 
	1.9 K
	68 hours

	5
	TBC2
	MBBL2086 
	[image: image35.png]



	11-09 
	22 h.
	32 h.[image: image36.png]



	-
	-
	-
	Prep 5 Pumping + Cool Down
	12-09
(07 : 06) 
	0 A. 
	Cooling
	54 hours

	6
	TBD1
	SSS043 
	[image: image37.png]



	01-09 
	52 h.
	63 h.
	121 h.
	5 h.
	43 h.
	Finished Ready to go Final scan
	13-09
(14 : 07) 
	0 A. 
	300 K
	285 hours

	7
	TBD2
	SSS046 
	[image: image38.png]



	07-09 
	138 h.[image: image39.png]



	-
	-
	-
	-
	Prep 5 Pumping + Cool Down
	12-09
(01 : 30) 
	0 A. 
	300 K
	138 hours

	8
	TBE1
	MBBL1132 
	[image: image40.png]



	11-09 
	12 h.
	48 h.[image: image41.png]



	-
	-
	-
	Prep 5 Pumping + Cool Down
	11-09
(15 : 35) 
	0 A. 
	Cooling
	60 hours

	9
	TBE2
	MBBL1144 
	[image: image42.png]



	09-09 
	20 h.
	50 h.
	21 h.
	2 h.[image: image43.png]



	-
	PT 12 Warm Up
	13-09
(13 : 36) 
	2 A. 
	Warming
	93 hours

	10
	TBF1
	MBBL1143 
	[image: image44.png]



	07-09 
	33 h.
	37 h.
	44 h.
	17 h.
	1 h.
	ICS 6 Disconnect Magnet WP07
	13-09
(03 : 38) 
	0 A. 
	300 K
	144 hours

	11
	TBF2
	MBBL1125 
	[image: image45.png]



	09-09 
	20 h.
	47 h.
	35 h.
	-5 h.[image: image46.png]



	-
	PT 12 Warm Up
	13-09
(20 : 30) 
	0 A. 
	Warming
	97 hours


Quench Information
 

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBAL3208
	[image: image47.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBAL3203
	12151 Amp.
(D1-L)
	12852 Amp.
([image: image48.jpg]


)
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	MBAR1140
	[image: image49.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	MBAL3205
	10996 Amp.
(D1-U)
	11491 Amp.
(D2_U)
	12499 Amp.
(D2-L)
	0 Amp.
(No Data)
	[image: image50.jpg]



	-
	-
	-
	-
	-
	-
	-
	-

	5
	MBBL2086
	[image: image51.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	SSS043
	11632 Amp.
(QE-34)
	12315 Amp.
(No Data)
	12339 Amp.
(QI-34)
	12708 Amp.
(E_34_S06)
	12856 Amp.
([image: image52.jpg]


)
	-
	-
	-
	-
	-
	-
	-
	-

	7
	SSS046
	[image: image53.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	8
	MBBL1132
	[image: image54.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	MBBL1144
	11540 Amp.
(D1-U)
	12517 Amp.
(D1-L)
	[image: image55.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	10
	MBBL1143
	11606 Amp.
(D1-L-S13)
	11858 Amp.
(D1-L)
	12199 Amp.
(D2-L s01)
	12347 Amp.
(D1 L S11)
	12535 Amp.
(D1 L S11)
	12416 Amp.
(D1-L_S11)
	[image: image56.png]



	-
	-
	-
	-
	-
	-

	11
	MBBL1125
	12506 Amp.
(D1-L)
	12406 Amp.
(D1-U)
	[image: image57.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-


General Problems last week :  
	Date
	Bench
	Problem
	Type
	Application/soft
	Action/advise

	Tuesday _07_A
	TBE2
	Quench data not saved in hf..


	AB/CO
	hf
	Guy informed. Giorgio analyse the quench using lf file. Switch off and on the DAQ hardware.

	Tuesday _07_N
	TBD1
	Tcoil: Reistance did not match.


	MTM
	
	Guy informed. Shaft not scanned No sticker for scanning..

	Tuesday _07_N
	TBD2
	In the connection box, one channel not connected D21-A2.. 


	MTM
	
	Corrected by Jackie.



	Tuesday _07_N

Wednesday _08_A
	TBF2

TBE2

TBC1&TBD1
	Quench data file not getting uploaded into data storage.

Idem

Idem.
	AB/CO
	Hf, lf
	

	Wednesday _08_M
	TBD1
	3 sectors not working (16-17, 00-01)


	MTM
	
	OK to continue like this, it will be repaired after cold test.

	Wednesday _08_A
	TBD1
	Load-line failed

Test repeated: failed again because de-ramp of current.
	?
	
	

	Wednesday _08_A
	TBE1
	Interlock on QH for E1: Qheaters 2 &4 not present.
	MTM
	
	Faulty contactor in the switch box.

Switch fron X2 to X1 then OK.

	Wednesday _08_A
	TBB1
	SSW measurement : wire broke
	MTM
	
	.

	Wednesday _08_N
	TBD1
	Not possible to power the MQT. Trigger in ext. auxiliary
	SL/PO
	
	Guy informed.Investigation : Probably coming from connection of Vtaps on clamp. Test skipped.

	Wednesday _08_N
	TBD1
	Cross calibration measurements blocked: Error in CPU address. Origin : Motor touching the end stop.
	AB/CO
	
	

	Thursday_09_A
	TBC2
	Provoked quench with LF heaters at 1.5 kA. Alarm on tfire.
	MTM
	
	Origin ?

Problem with QH time scaling: Fixed by Boris.

	Thursday_09_N
	TBB2
	SPA on dipole 3206:launched 4 times. It stopped at 60 A and current ramp down.

Setting SSS loaded instead of dipole
	AB/CO? IT?
	
	Correct setting loaded then OK.

	Thursday_09_N
	TBD1
	4 SPA on D bench.
	AB/CO? IT?
	
	

	Friday_10_A
	TBA1

TBF1
	Qloc compensation problem

Qloc compensation problem
	MTM
	
	Guy informed D20 connector not connected (TBA1)!

	Friday_10_A
	TBF1
	Interlock Ext2 on PC.
	MTM
	
	Manual valves of water supply of PC closed.

	Friday_10_N
	TBD2
	Anticryostat problem (long, aperture 1)  in SSS.
	MTM
	
	One resistance shorted to the ground (the heating one). Guy informed. OK given to continue. Investigation planned during the disconnection.

	Saturday_11_M
	TBD1
	CDAP at 1.9 K : HV problem for orbit correctors
	MTM
	
	Guy informed : test re-launched then OK.

	Saturday_11_N
	TBD1
	SSW measurements with system 1 : problem with the tension imposed by the motor:.
	MTM
	
	Guy informed. He found that the new controller needs a special configuration. Guy exchange with system 2. Joe Dimarco will be contacted.

	Sunday_12_M
	TBF1
	On 1143 CDAP OK on sun station but not in the generated report.
	MTM
	HV
	Guy will inform Richard Monday.

	Sunday_12_M
	TBC1
	3205: First training quench: No data transferred : analyse done locally
	AB/CO
	Hf-lf
	








